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Allen, Pam, NMENV ' ~,' ~i'~~,.t~J .. 4 

From: Maestas, Ricardo, NMENV 
Sent: Thursday, December 18, 2014 9:47 AM 

Allen, Pam, NMENV To: 
Subject: FW: WIPP Information - No Call Today 

From: Oba Vincent [mailto:oba.vincent@cbfo.doe.gov] 
Sent: Wednesday, May 28, 2014 9:17 AM 
To: Kliphuis, Trais, NMENV; 'peake.tom@epa.gov'; 'Edwards, Jonathan'; 'Walsh, Jonathan'; 'Perrin, Alan'; 'Kehrman, Bob -
RES'; 'Chavez, Rick - RES'; 'Stone.Nick@epa.gov'; Smith, Coleman, NMENV; 'brozowski.george@epa.gov'; 'Fraass, Ron'; 
'Hardy, Russell'; 'Veal.Lee@epamail.epa.gov'; 'Economy, Kathleen (Economy.Kathleen@epa.gov)'; 'Poppell, Sam W. 
(Poppell.Sam@epa.gov)'; Maestas, Ricardo, NMENV; 'Faller, Scott H.' 
Cc: George Basabilvazo - WIPPNet; 'Reynolds, Tammy - NWP'; 'Harris, Alton - DOE EM'; Susan McCauslin; 'Joe Harvill 
(iharvill@portaqeinc.com)'; 'Kennedy, Scott - NWP'; 'Jones, Stewart - RES'; 'Oates, Berta - CTAC'; 
'schultheisz.daniel@epa.gov'; Philip Theisen - ORISE; Russ Patterson - WIPPNet; 'Kouba, Steve - WRES'; Roger Nelson -
WIPPNet; 'Bignell, Dale - CTAC'; Susan McCauslin - WIPPNet; 'Pace, Berry'; Anthony Stone - WIPPNet; J.R. Stroble -
WIPPNet; 'Lynnes, Kate'; Heidi Lowe; 'skeffer@projectenhancement.com' 
Subject: WIPP Information - No Call Today 

Below is a summary of activities: (Please note, activities and dates are subject to change. Please verify the most 
current dates of any information provided). 

• 5/28/14 entry: No entry to the waste face will be made during this entry. The emphasis will be the adjusting of 

the U/G ventilation and performing ground control surveys. Personnel will transverse down E140 to 53080. 

• Planned 5/29/14 entry: Additional investigation of the area of concern with an emphasis on sample collection of 

the material from the damaged drum and MgO. The plans are to collect samples of MgO at 6 locations, material 

on top of SWB, and material at the mouth of the breached drum. The CAM data will be downloaded. Limited 

video (in support of the sampling) will be obtained. 

• Filter replacement has been postponed to allow for additional underground entries. Construction of the 

containment tents required for the activity is complete. Installation of the make-up air inlet into the ductwork 

and ventilation configuration changes are being rescheduled as well. The tentative date is presently 5/30. 

• The preliminary analytical report from SRS for the U/G samples has been posted to the portal. 

Thank you 

Oba 
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